

LUND UNIVERSITY

Chromosome x-wide association study identifies Loci for fasting insulin and height and evidence for incomplete dosage compensation.

Tukiainen, Taru; Pirinen, Matti; Sarin, Antti-Pekka; Ladenvall, Claes; Kettunen, Johannes; Lehtimäki, Terho; Lokki, Marja-Liisa; Perola, Markus; Sinisalo, Juha; Vlachopoulou, Efthymia; Eriksson, Johan G; Groop, Leif; Jula, Antti; Järvelin, Marjo-Riitta; Raitakari, Olli T; Salomaa, Veikko; Ripatti, Samuli

Published in: **PLoS Genetics**

DOI: 10.1371/journal.pgen.1004127

2014

Link to publication

Citation for published version (APA):

Tukiainen, T., Pirinen, M., Sarin, A.-P., Ladenvall, C., Kettunen, J., Lehtimäki, T., Lokki, M.-L., Perola, M., Sinisalo, J., Vlachopoulou, E., Eriksson, J. G., Groop, L., Jula, A., Järvelin, M.-R., Raitakari, O. T., Salomaa, V., & Ripatti, S. (2014). Chromosome x-wide association study identifies Loci for fasting insulin and height and evidence for incomplete dosage compensation. PLoS Genetics, 10(2), Article e1004127. https://doi.org/10.1371/journal.pgen.1004127

Total number of authors:

General rights

Unless other specific re-use rights are stated the following general rights apply:

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal

Read more about Creative commons licenses: https://creativecommons.org/licenses/

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

LUND UNIVERSITY

PO Box 117 221 00 Lund +46 46-222 00 00

Chromosome X-Wide Association Study Identifies Loci for Fasting Insulin and Height and Evidence for Incomplete Dosage Compensation

Taru Tukiainen^{1,2,3}, Matti Pirinen¹, Antti-Pekka Sarin^{1,4}, Claes Ladenvall⁵, Johannes Kettunen^{1,4}, Terho Lehtimäki⁶, Marja-Liisa Lokki⁷, Markus Perola^{1,8,9}, Juha Sinisalo¹⁰, Efthymia Vlachopoulou⁷, Johan G. Eriksson^{8,11,12,13,14}, Leif Groop^{1,5}, Antti Jula¹⁵, Marjo-Riitta Järvelin^{16,17,18,19,20}, Olli T. Raitakari^{21,22}, Veikko Salomaa⁸, Samuli Ripatti^{1,4,23,24}*

1 Institute for Molecular Medicine Finland (FIMM), University of Helsinki, Helsinki, Finland, **2** Analytic and Translational Genetics Unit, Massachusetts General Hospital, Boston, Massachusetts, United States of America, **3** Program in Medical and Population Genetics, Broad Institute of Harvard and MIT, Cambridge, Massachusetts, United States of America, **4** Unit of Public Health Genomics, National Institute for Health and Welfare, Helsinki, Finland, **5** Department of Clinical Sciences, Diabetes and Endocrinology, Lund University and Lund University Diabetes Centre, CRC at Skåne University Hospital, Malmö, Sweden, **6** Department of Clinical Chemistry, Fimlab Laboratories, University of Tampere School of Medicine, Tampere, Finland, **7** Transplantation Laboratory, Haartman Institute, University of Tartu, Tartu, Estonia, **10** Division of Cardiology, Department of Medicine, Helsinki University Central Hospital, Helsinki, Finland, **11** Department of General Practice and Primary Healthcare, University of Helsinki, Helsinki, Finland, **12** Unit of General Practice, Helsinki University Central Hospital, Helsinki, Finland, **13** Folkhälsan Research Center, Helsinki, Finland, **16** Department of Epidemiology and Biostatistics, MRC Health Protection Agency (HPA) Centre for Environment and Health, School of Public Health, Imperial College London, London, United Kingdom, **17** Institute of Health Sciences, University of Oulu, Oulu, Finland, **18** Biocenter Oulu, University of Oulu, Oulu, Finland, **21** Department of Clinical Physiology and Nuclear Medicine, University of Turku and Turku University Hospital, Turku, Finland, **22** Research Centre of Applied and Preventive Cardiovascular Medicine, University of Turku, Turku, Finland, **23** Wellcome Trust Sanger Institute, Hinxton, Cambridge, United Kingdom, **24** Hjelt Institute, University of Helsinki, Finland

Abstract

The X chromosome (chrX) represents one potential source for the "missing heritability" for complex phenotypes, which thus far has remained underanalyzed in genome-wide association studies (GWAS). Here we demonstrate the benefits of including chrX in GWAS by assessing the contribution of 404,862 chrX SNPs to levels of twelve commonly studied cardiometabolic and anthropometric traits in 19,697 Finnish and Swedish individuals with replication data on 5,032 additional Finns. By using a linear mixed model, we estimate that on average 2.6% of the additive genetic variance in these twelve traits is attributable to chrX, this being in proportion to the number of SNPs in the chromosome. In a chrX-wide association analysis, we identify three novel loci: two for height (rs182838724 near *FGF16/ATRX/MAGT1*, joint P-value = 2.71×10^{-9} , and rs1751138 near *ITM2A*, P-value = 3.03×10^{-10}) and one for fasting insulin (rs139163435 in Xq23, P-value = 5.18×10^{-9}). Further, we find that effect sizes for variants near *ITM2A*, a gene implicated in cartilage development, show evidence for a lack of dosage compensation. This observation is further supported by a sex-difference in *ITM2A* expression in whole blood (P-value = 0.00251), and is also in agreement with a previous report showing *ITM2A* escapes from X chromosome inactivation (XCI) in the majority of women. Hence, our results show one of the first links between phenotypic variation in a population sample and an XCI-escaping locus and pinpoint *ITM2A* as a potential contributor to the sexual dimorphism in height. In conclusion, our study provides a clear motivation for including chrX in large-scale genetic studies of complex diseases and traits.

Citation: Tukiainen T, Pirinen M, Sarin A-P, Ladenvall C, Kettunen J, et al. (2014) Chromosome X-Wide Association Study Identifies Loci for Fasting Insulin and Height and Evidence for Incomplete Dosage Compensation. PLoS Genet 10(2): e1004127. doi:10.1371/journal.pgen.1004127

Editor: Peter M. Visscher, The University of Queensland, Australia

Received June 4, 2013; Accepted December 5, 2013; Published February 6, 2014

Copyright: © 2014 Tukiainen et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This research was supported through funds from The European Community's Seventh Framework Programme (FP7/2007-2013), EU-BioSHaRE Consortium (grant agreement 261433) (https://www.bioshare.eu). SR was supported by the Academy of Finland Center of Excellence in Complex Disease Genetics (no.s 213506 and 129680) (http://www.aka.fi/eng), Academy of Finland (no. 251217) (http://www.aka.fi/eng), the Finnish foundation for Cardiovascular Research (http://www.sydantutkimussaatio.fi) and the Sigrid Juselius Foundation (http://www.sigridjuselius.fi/foundation). MPi was funded by the Academy of Finland (257654) (http://www.aka.fi/eng). VS was supported by the Academy of Finland (139635) (http://www.aka.fi/eng), and the Finnish Foundation for Cardiovascular Research (http://www.sydantutkimussaatio.fi). TT received funding from Emil Aaltonen Foundation (http://www.emilaaltonen.fi). MPe is partly financially supported for this work by the Finnish Academy SALVE program "Pubgensense" 129322 (http://www.aka.fi/eng) and by grants from Finnish Foundation for Cardiovascular Research (http://www.sydantutkimussaatio.fi). The Young Finns Study has been financially supported by the Academy of Finland (http://www.aka. fi/eng): grants 134309 (Eye), 126925, 121584, 124282, 129378 (Salve), 117787 (Gendi), and 41071 (Skidi), the Social Insurance Institution of Finland (http://www. kela.fi/web/en), Kuopio, Tampere and Turku University Hospital Medical Funds (grant 9N035 for TL), Juho Vainio Foundation (http://www.juhovainionsaatio.fi/ pages/in-english/home.php), Paavo Nurmi Foundation (http://www.paavonurmensaatio.fi/saatio_e3.htm), Finnish Foundation of Cardiovascular Research (http:// www.sydantutkimussaatio.fi) and Finnish Cultural Foundation (http://www.skr.fi/en), Tampere Tuberculosis Foundation and Emil Aaltonen Foundation (http:// www.emilaaltonen.fi). The Helsinki Birth Cohort Study (HBCS) has been supported by grants from the Academy of Finland (http://www.aka.fi/eng), the Finnish Diabetes Research Society (http://www.diabetes.fi/en/finnish_diabetes_association), Folkhälsan Research Foundation (http://www.folkhalsan.fi/en/startsida/ About-Folkhalsan/), Novo Nordisk Foundation (http://www.novonordiskfonden.dk/en), Liv och Hälsa (http://www.livochhalsa.fi), Finska Läkaresällskapet (http:// www.fls.fi/sallskapet/), Signe and Ane Gyllenberg Foundation (http://gyllenbergs.fi/en/signe-och-ane-gyllenbergs-stiftelse-en/), Ahokas Foundation and Juho Vainio Foundation (http://www.juhovainionsaatio.fi/pages/in-english/home.php). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Introduction

Genome-wide association studies (GWAS) have discovered a wealth of loci associated with complex phenotypes with almost 5,800 significant associations for more than 500 different phenotypes reported in the NHGRI GWAS catalog [1] (accessed August 13, 2013). These GWAS discoveries are, however, concentrated on the autosomes leaving the sex chromosomes, especially the relatively large X chromosome (chrX), underrepresented; while chrX contains approximately 5% of genomic DNA, hence being comparable in size to chromosome 7, and encodes for more than 1,500 genes, only around 20 unique significantly associated X-chromosomal loci in total are recorded in the catalog. For instance, there are hundreds of known autosomal loci for height, BMI and blood lipids, but only one significant height locus has been identified in chrX, and this in individuals of African ancestry, and no X-chromosomal associations for these other highly polygenic phenotypes have been reported. Nevertheless, almost 1% of genetic variance in height and BMI has been shown to be accountable to chrX SNPs [2], demonstrating that common genetic variation in chrX contributes to complex phenotypes.

A likely explanation for the dearth of association findings in chrX is that the chromosome is often neglected in GWAS: Wise et al. recently surveyed all published GWAS from 2010 and 2011 and found that only 33% of these studies had included chrX analyses [3]. While some association studies have opted for including chrX, such as recent genetic screens on sex-hormone binding globulin levels [4] and Grave's disease [5], removal of non-autosomal data appears to be a common practice in GWAS [6,7]. There are many potential reasons for the exclusion of chrX in GWAS, as outlined by Wise et al [3], a major contributor being that the analysis pipeline applied for autosomes is not directly applicable to chrX analyses.

While women carry two copies of chrX, men are hemizygous for the chromosome. The allele dosages between the sexes are balanced by random X chromosome inactivation (XCI) that silences one of the two chromosomes in women, hence requiring the allele coding for chrX markers to be adjusted accordingly for the analyses. However, XCI does not evenly cover the whole of chrX, but approximately 15% of the loci in the chromosome completely escape from XCI and in further 10% of the sites the silenced chrX is variably active in women, although the expression from the inactivated copy of chrX is often lower than from the active chrX [8]. The incomplete XCI adds another layer of analytical challenges, yet at the same time it also makes chrX particularly interesting to study, as the regions of incomplete dosage compensation are among the genomic contributors to the differences between gene dosages in men and women. As such, these loci could partly explain phenotypic sexual dimorphisms and additionally contribute to the phenotypic characteristics observed in chrX aneuploidies.

Given the underutilization of chrX data in previous studies and hence the potential for novel biological discoveries, we aimed at surveying the contribution of the chromosome to complex traits. To this end, we expanded the marker set for chrX by imputing the non-pseudoautosomal region of chrX in almost 25,000 Finnish and Swedish individuals from seven discovery and one replication cohort (Table 1) by utilizing the recently released comprehensive reference panel from the 1000 Genomes Project [9]. We focused our chrX-wide screen on twelve quantitative anthropometric and

cardiometabolic phenotypes for which hundreds of autosomal, but no X-chromosomal, loci have been identified in GWAS of individuals of European ancestry, namely height, body-mass-index (BMI), waist-hip-ratio (WHR), systolic and diastolic blood pressure (SPB and DBP), C-reactive protein, fasting insulin and glucose, total, LDL and HDL cholesterol (TC, LDL-C and HDL-C) and triglycerides (TG). By using a linear mixed model, we show that the variation in chrX influences the levels of many of these complex phenotypes and in an association analysis identify and replicate three new associated X-chromosomal loci, one for fasting insulin and two for height, hence demonstrating the value of assessing chrX associations. Further, we find strong evidence for a lack of dosage compensation in one of the two associated height loci by applying a meta-analysis that allows for sex heterogeneity in effects and by a formal statistical model comparison between the different dosage compensation models given the observed data.

Results

Genetic variance in chrX contributes to the levels of many anthropometric and metabolic phenotypes

We first estimated the proportion of variance in each of the twelve phenotypes accountable jointly to the common and lowfrequency (minor allele frequency (MAF) >1%) SNPs in chrX using a linear mixed model [2,10] in the study samples for which the individual-level genotype data were available (Materials and Methods). ChrX variants were estimated to contribute to the levels of many of these phenotypes (Table 2): under the model of equal variance in males and females (see Materials and Methods for discussion about the models), more than 0.5% of the variance in height, SBP, HDL-C, fasting glucose and insulin appear to be due to chrX variation, hence motivating the search for associated variants in chrX. The highest estimate for X-linked variance $(1.4\%, \text{ P-value} = 2.00 \times 10^{-6})$ was observed for height, a highly heritable and polygenic phenotype. For the other phenotypes the statistical significance of the estimates was not overwhelming, a result of the available sample size and lower trait heritability, but also the estimates for SBP, HDL-C and fasting insulin were significantly different (P-value<0.05) from zero. In these four phenotypes, which showed non-zero X-linked variance, on average 4.0% (range 2.4%-6.0%) of the estimated whole-genome genetic variance was attributable to chrX while the corresponding value over all twelve traits was 2.6%.

Following the work of Yang et al. [2] we calculated the variance estimates under three different models for dosage compensation, i.e., equal variance between men and women, full dosage compensation and no dosage compensation (Table S1). The differences between the model fit were small (as measured by likelihood-ratios) and none of the models was consistently favored above the other two. This is likely due to our sample size being limited for such comparisons, but may also reflect the differences in the genetic architecture of the various loci in chrX that contribute to the variance for each phenotype.

ChrX-wide association analysis identifies three associated loci

In order to identify X-linked loci contributing to the phenotypic variance we assessed associations between directly genotyped and imputed chrX SNPs and the twelve phenotypes across seven discovery cohorts (N = 19,697; Table 1). As the majority of the loci

Author Summary

The X chromosome (chrX) analyses have often been neglected in large-scale genome-wide association studies. Given that chrX contains a considerable proportion of DNA, we wanted to examine how the variation in the chromosome contributes to commonly studied phenotypes. To this end, we studied the associations of over 400,000 chrX variants with twelve complex phenotypes, such as height, in almost 25,000 Northern European individuals. Demonstrating the value of assessing chrX associations, we found that as a whole the variation in the chromosome influences the levels of many of these phenotypes and further identified three new genomic regions where the variants associate with height or fasting insulin levels. In one of these three associated regions, the region near ITM2A, we observed that there is a sex difference in the genetic effects on height in a manner consistent with a lack of dosage compensation in this locus. Further supporting this observation, ITM2A has been shown to be among those chrX genes where the X chromosome inactivation is incomplete. Identifying phenotype associations in regions like this where chrX allele dosages are not balanced between men and women can be particularly valuable in helping us to understand why some characteristics differ between sexes.

in chrX are subject to XCI, we adopted an allele coding that is consistent with the full dosage compensation model, i.e. we treated hemizygous men as equivalent to homozygous women (Materials and Methods). Within each cohort the associations were first studied separately for males and females using SNPTEST [11] and the results were subsequently combined in a fixed-effects metaanalysis in GWAMA [12]. By analyzing the associations of up to 405,411 polymorphic high-quality SNPs we identified three associated loci (P-value $<5 \times 10^{-8}$): two for height (both in Xq21.1) and one for fasting insulin (in Xq23) (Table 3, Table S2, Figure 1). We followed-up these findings in an independent replication set and found further evidence for association in all three loci (discovery and replication combined up to N = 24,729), with all lead SNP P-values $<6 \times 10^{-9}$ (Table 3).

In the more strongly associated height locus the associated SNPs (lead SNP rs1751138, joint, i.e., discovery and replication combined, P-value = 3.03×10^{-10} , MAF = 0.36) are located approximately 35 kb upstream of ITM2A (integral membrane protein 2A), a gene implicated in early cartilage development [13,14]. We observed that the minor A allele of rs1751138, which is associated with shorter stature, is also associated with an increased expression of ITM2A in whole blood (P-val $ue = 6.23 \times 10^{-14}$, N = 513; Materials and Methods), providing further evidence for ITM2A as a functional candidate gene for this association. The second region associated with height spans FGF16, ATRX and MAGT1. The lead SNP (rs182838724, joint Pvalue = 2.71×10^{-9} , MAF = 0.30) is intronic within *ATRX*, a gene associated with the X-linked alpha thalassaemia mental retardation syndrome (ATR-X), a rare condition manifesting itself as profound developmental delay often accompanied by several other distinct characteristics including skeletal abnormalities in 90% and short stature in two thirds of the affected individuals [15]. As the two height lead SNPs map only 2 Mb apart, we confirmed that the associations are independent of each other by conditioning the association analysis on the lead SNP of the ITM2A locus. The conditional analysis did not attenuate the height signal in the ATRX region, yet here the most associated SNP (rs34979608, joint P-value = 1.52×10^{-9} , r² with rs182838724 = 0.91; Table S3, Figure S1) maps outside ATRX, 4 kb downstream of MAGT1, a gene encoding a magnesium transporter. Both of the height associations are present already in childhood (P-value for

Table 1. A summary of the characteristics of the discovery and replication cohorts.

Cohort	Full name	Sex	Ν	Age (years)	Height (cm)	BMI (kg/m²)
Discovery						
NFBC	Northern Finland Birth Cohort 1966	Males	2388	31.0±0.0	178.2±6.4	25.2±3.6
		Females	2644	31.0±0.0	164.8±6.2	24.2±4.7
COROGENE	The COROGENE Study	Males	2441	60.4±12.8	176.0±6.7	27.4±4.2
		Females	1502	62.8±13.4	161.6±6.6	26.9±5.2
DGI	Diabetes Genetics Initiative	Males	1534	61.0±10.6	174.9±6.3	27.5±3.7
		Females	1608	62.5±10.7	161.7±6.1	27.8±4.7
GENMETS	Health2000 GenMets Study	Males	1000	49.2±10.4	176.4±6.6	27.3±3.9
		Females	1040	52.1±11.4	162.7±6.5	27.2±5.0
YFS	The Cardiovascular Risk in Young Finns Study	Males	917	37.6±5.1	179.7±6.7	26.8±4.3
		Females	1111	37.6±5.0	166.0±6.0	25.3±5.0
PredictCVD	Case control sample from the FINRISK surveys	Males	1180	51.5±13.0	174.5±6.8	27.6±4.3
		Females	686	52.2±13.4	161.5±6.5	27.3±5.3
HBCS	Helsinki Birth Cohort Study	Males	696	61.4±2.8	176.9±5.8	27.5±4.3
		Females	950	61.5±3.0	163.2±5.8	27.7±5.1
Replication						
FINRISK	Subset from the FINRISK 1997 and 2002 surveys	Males	2287	46.8±13.4	175.8±7.0	26.7±4.1
		Females	2745	45.1±12.4	162.6±6.2	26.2±5.1

N: maximum number of individuals with phenotype and genotype data available; BMI: Body-mass-index; Age, height and BMI are given as mean ± standard deviation. doi:10.1371/journal.pgen.1004127.t001

Table 2. Estimates of the explained variances in the twelvequantitative phenotypes attributable to chromosome X SNPsand autosomal SNPs separately using equal variance (EV)model.

Phenotype	N	h _x (%)	se _x (%)	P-value	h _{aut} (%)	se _{aut} (%)
Height	14408	1.41	0.41	2.00E-06	52.35	2.41
SBP	9990	1.07	0.52	0.005	16.63	2.98
Fasting glucose	9151	0.84	0.57	0.06	11.15	3.09
HDL-C	11139	0.73	0.42	0.01	30.21	2.85
Fasting insulin	9616	0.68	0.47	0.04	12.66	2.98
TG	11140	0.43	0.4	0.1	19.88	2.72
CRP	9697	0.42	0.47	0.2	11.16	2.89
WHR	12334	0.22	0.34	0.2	12.75	2.36
BMI	14214	0.11	0.31	0.4	25.86	2.25
DBP	9984	0	0.42	0.5	12.5	2.89
LDL-C	11040	0	0.41	0.5	26.43	2.82
тс	11141	0	0.43	0.5	27.21	2.81

The estimates are based on an analysis of the individuals from six Finnish cohorts using the program GCTA and 217,112 common and low-frequency chrX SNPs (MAF>1%) directly genotyped or imputed with high-quality (info >0.8) and 319,445 directly genotyped autosomal SNPs (MAF>1%).

h_x: estimate for the proportion of explained variance accountable by the SNPs in chromosome X in per cent; se_x: standard error in per cent for the X chromosome variance estimate; P-value: P-value for the test of $h_x = 0$; h_{aut} : estimate for the proportion of explained variance accountable by the SNPs in autosomes in per cent; se_{aut}: standard error in per cent for the autosomal variance estimate; SBP: systolic blood pressure; HDL-C: high-density lipoprotein cholesterol; TG: total triglycerides; CRP: C-reactive protein; WHR: waist-hip-ratio; BMI: body-mass-index; DBP: diastolic blood pressure; LDL-C: low-density lipoprotein cholesterol; TC: total cholesterol.

doi:10.1371/journal.pgen.1004127.t002

ITM2A = 1.58×10^{-5} , P-value for *ATRX* = 0.00955, N = 3287, subset of two of the study cohorts, ages 8–10; Table S4), but for the *ATRX* locus the association appears weaker in children than in the same individuals in adulthood (beta in childhood = 0.059, beta in adulthood = 0.092, P-value for difference in effect sizes = 0.043; Table S4) suggesting additional influence of puberty. In the third associated locus, the dosage of the minor G allele of rs139163435 (MAF = 0.071) was robustly associated with lower levels of fasting insulin across all cohorts (joint P-value = 5.18×10^{-9}). The lead SNP maps to an apparent gene desert: the gene closest to the association, *SLC6A14*, a possible candidate gene for X-linked obesity (OMIM: 300444), lies more than 500 kb away.

Genetic effects near ITM2A suggest a lack of dosage compensation

Up to 25% of the chrX loci may not be subject to complete dosage compensation [8], but in these regions also the inactivated X chromosomes are transcriptionally fully or partially active. When adopting an allele coding that is consistent with the full dosage compensation model, as we did in the cohort-level analyses, the genetic effects in these incompletely dosage compensated loci are expected to be larger in absolute value in women than in men. The fixed-effects meta-analysis does not, however, account for potential sex heterogeneity. Therefore, we complemented the fixed-effects analysis by performing a metaanalysis that treats the male and female-specific genetic effects separately, a so-called sex-differentiated meta-analysis (Materials and Methods), in order to indicate loci showing incomplete dosage compensation and to potentially also facilitate the discovery of new associations.

Allowing for different effect sizes between males and females in the meta-analysis pinpointed no further loci, yet the ITM2A lead height SNP was more strongly associated in this sex-differentiated analysis (joint P-value = 3.26×10^{-12} ; Table 3). Pointing to lack of dosage compensation, in this locus the allelic effects were estimated to be more than twice the size in women compared to men when coding hemizygous men equal to homozygous women (standardized beta in females: 0.093, se: 0.014; beta in males: 0.035, se: 0.009; P-value for the difference in the male and female effects = 2.85×10^{-4} ; Table 3). For the other two new loci there was no indication of heterogeneity in the male and female effects (Table 3). Accordingly, the proportion of variance explained was approximately twice the size in men compared to women in the ATRX (0.22% vs. 0.14% for height) and Xq23 (0.51% vs. 0.19% for fasting insulin) loci, as expected under the model of random XCI, but not for the ITM2A SNP (0.12% vs. 0.39% for height) (explained variances calculated under the assumption of full dosage compensation; Table 3; Materials and Methods). The observed deviation from the full dosage compensation model in the ITM2A locus was not driven by differences in allele frequencies or sample sizes, as these are similar between men and women (Tables 1 and 3).

Building up evidence for a lack of dosage compensation in the ITM2A locus

To verify the observation of lack of dosage compensation in the ITM2A locus for height, we formally investigated how the models of full dosage compensation and no dosage compensation explain the data in the three associated loci pinpointed in our chrX-wide association analysis. To this end we devised a novel application of a Bayesian model-comparison framework [16] (Materials and Methods), to quantify the relative support for each of the dosage compensation models. While the Xq23 and ATRX associations were highly consistent with full dosage compensation there was clear evidence towards complete escape from the inactivation in the ITM2A locus: Assuming that both models are equally probable *a priori*, the posterior probabilities for the no dosage compensation model are 0.07 for rs13916345 in Xq23 (fasting insulin), and 0.18 for rs182838724 near ATRX (height), but 0.99 for rs1751138 near ITM2A (height) (Figure 2).

As lack of dosage compensation in the ITM2A locus in women should be reflected in the level of ITM2A expression, given the two actively transcribed X chromosomes, we evaluated the sex difference in the level of the ITM2A expression probe that had earlier showed a significant *cis*-effect with the lead SNP of the locus. The average level of ITM2A expression in whole blood was observed to be higher in women (P-value = 0.00251; Figure S2; Materials and Methods), thus providing further support for incomplete XCI in this locus.

Discussion

Motivated by the underrepresentation of reported GWAS discoveries in chrX, we investigated the association of chrX to twelve anthropometric and cardiometabolic traits in more than 24,500 individuals using a high-resolution map of non-pseudoautosomal chrX SNPs. Our data demonstrate that SNPs in chrX are associated with many of the studied phenotypes, including the three novel loci pinpointed in our chrX-wide association analysis. Additionally, our discovery of lack of dosage compensation for height near *ITM2A* not only highlights the value of accounting for potential sex heterogeneity when assessing chrX associations, but

Locus/Candidates	SNP	Pos (chrX)	EA/OA	Data set	Sex	EAF	Beta	SE	P-value	z	P-value (se)	x) P-value (het)	Variance explained (%)
Xq21.1/ITM2A	rs1751138	78657806	G/A	Discovery	Males+Females	0.643	0.057	0.009	5.54E-10	19566	1.99E-10	1.33E-02	
					Males	0.641	0.045	0.011	3.32E-05	10093		I	0.19%
					Females	0.644	0.084	0.016	1.67E-07	9473		I	0.32%
				Replication	Males+Females	0.647	0.038	0.019	5.53E-02	4996	2.53E-04	3.32E-04	1
					Males	0.647	-0.013	0.024	5.94E-01	2259		I	0.01%
					Females	0.647	0.128	0.032	5.51E-05	2737		I	0.75%
				Joint	Males+Females	0.643	0.054	0.008	3.03E-10	24562	3.26E-12	2.85E-04	
					Males	0.642	0.036	0.010	3.39E-04	12352	,	I	0.12%
					Females	0.645	0.093	0.014	2.56E-10	12210		I	0.39%
Xq21.1/FGF16, ATRX, MAGT	1 rs182838724	76797439	T/A	Discovery	Males+Females	0.299	0.053	0.009	4.31E-08	19562	1.31E-07	1.99E-01	1
					Males	0.294	0.053	0.011	4.07E-06	10092		I	0.23%
					Females	0.304	0.054	0.016	1.24E-03	9470	,	I	0.12%
				Replication	Males+Females	0.290	0.056	0.020	7.74E-03	4996	1.69E-02	3.03E-01	1
					Males	0.287	0.046	0.025	6.60E-02	2259	,	I	0.18%
					Females	0.293	0.071	0.033	2.88E-02	2737		I	0.21%
				Joint	Males+Females	0.297	0.054	0.008	2.71E-09	24558	4.44E-09	8.19E-02	1
					Males	0.293	0.052	0.010	8.95E-07	12351		I	0.22%
					Females	0.301	0.058	0.015	1.58E-04	12207	,	I	0.14%
Xq23	rs139163435	116110239	G/T	Discovery	Males+Females	0.071	-0.128	0.023	2.87E-08	11681	1.72E-07	5.74E-01	,
					Males	0.071	-0.137	0.028	1.45E-06	5542	,	I	0.50%
					Females	0.071	-0.110	0.039	4.95E-03	6139		I	0.16%
				Replication	Males+Females	0.088	-0.221	0.110	4.51E-02	370	9.95E-02	4.39E-01	ı
					Males	0.091	-0.170	0.129	1.87E-01	200		I	0.96%
					Females	0.084	-0.363	0.214	9.00E-02	170	,	I	2.04%
				Joint	Males+Females	0.072	-0.132	0.023	5.18E-09	12051	3.46E-08	6.65E-01	ı
					Males	0.072	-0.139	0.028	6.05E-07	5742	,	I	0.51%
					Females	0.071	-0.118	0.039	2.15E-03	6309		T	0.19%

Table 3. The lead associations in the three significantly associated loci in the chromosome X-wide association analysis.



Figure 1. A Manhattan plot across all the twelve phenotypes and regional association plots for the three associated loci. A. A Manhattan plot showing the associations of the X chromosome SNPs to the twelve phenotypes in the discovery analysis. The associated loci (P-value $<5.0 \times 10^{-8}$) are highlighted with red dots and solid lines. B. A plot of the height associations in the Xq21.1 region showing two separate association signals. C–E. The association plots for height near *ITM2A* (C) and height near *ATRX* (D) and for fasting insulin in Xq23 (E), showing the association P-values in the discovery analysis. Yellow diamonds indicate the SNPs, which showed the strongest evidence of association in each of the loci, and purple diamonds and the P-values given in the plots indicate the associations of these lead SNPs in the joint analysis of discovery and replication data sets. Each circle in the plots indicates a SNP with the color of the circle (in C–E) showing the linkage disequilibrium between the SNP and the highlighted lead SNP: dark blue (r^2 <0.2), light blue (r^2 >0.2), green (r^2 >0.4), orange (r^2 >0.6) and red (r^2 >0.8), The r^2 values were calculated using the genotype data from the COROGENE cohort, and the recombination rate, indicated by the blue lines in the background and the right hand y-axis, was estimated from the CEU HapMap data. The bottom panels show the genes (RefSeq Genes) and their positions in each locus. In all plots the dashed red line marks the threshold for genome-wide significance (P-value = 5.0×10^{-8}).



Figure 2. Comparison of the dosage compensation models in the three associated loci applying Bayesian framework. A. Separately estimated effect sizes of the three lead SNPs (dots labeled with rs-numbers) in females (x-axis) and males (y-axis) when female genotypes are coded {0,1,2} and male genotypes {0,2}. Ellipses show the 95% confidence regions for the estimates. The lines show the regions of the expected values of the effects under either full dosage compensation (FDC) or no dosage compensation (NDC) models. The associated traits are fasting insulin (INS) and height (HGT). B. Posterior probability of no dosage compensation (NDC) model at the three lead SNPs when the other candidate is full dosage compensation model and the two models are equally probable *a priori*. Labels under bars give the rs-number of the SNP, the associated trait (INS = fasting insulin or HGT = height) and the height of the bar. doi:10.1371/journal.pgen.1004127.g002

also manifests that some of the X-linked loci may contribute to sexual dimorphisms, in this case to the height difference between men and women.

The contribution of common X-chromosomal SNPs to a few complex phenotypes have been explored previously: Yang et al. showed that between 0.57% and 0.82% of the variance in height, BMI and von Willebrand factor is explained by the SNPs in chrX [2]. In our study, we extended the estimates to further ten anthropometric and cardiometabolic phenotypes, and also use a more comprehensive set of SNPs, and similarly demonstrate that a small but non-negligible proportion, up to 1.4%, of the total variance in many of the twelve phenotypes studied appears attributable to chrX. While the variance estimates for autosomes have been observed to be proportional to the chromosome length [2], our estimates for chrX were on average 2.6% of the total genetic variance estimate, and therefore below what would be expected based on the proportion of the genomic DNA contained within chrX, i.e., approximately 5%. However, given the smaller population size and lower mutation rate, chrX is genetically less diverse than the autosomes, and indeed our estimates appear to be more in line with around 3.4% of the SNPs in the 1000 Genomes reference being X-chromosomal [9] (Figure S3). This implies that around 3% of all GWAS discoveries could be hidden in chrX and hence with the inclusion of chrX in GWAS further X-chromosomal loci for complex traits will be discovered.

As thus far chrX has only infrequently been included in GWAS of the twelve traits studied, it is unsurprising that the three loci now discovered, one for fasting insulin and two for height, represent the first ones reported in chrX for these phenotypes in European populations. Two earlier GWAS conducted in individuals of African ancestry discovered a height locus in Xp22.3 [17,18], yet

the associated lead SNP is monomorphic in Europeans and the region shows no signal in our analyses. One of the early height GWAS including mainly European samples showed a suggestive association (P-value = 3×10^{-6}) [19] mapping 8.6 kb from the *ITM2A* association discovered in our analyses (r² = 0.728 between the lead SNPs), however, in that study the finding failed to replicate and thus never reached the formal threshold for significance.

While further studies are required to elaborate the causative variants underlying the association with fasting insulin in Xq23, in both of the height loci there are plausible candidate genes that could be responsible for the observations. The first height association spans FGF16, ATRX and MAGT1 in a gene rich region. Both ATRX and MAGT1 have been implicated in mental retardation syndromes, and interestingly the syndrome associated with mutations in ATRX is often accompanied with skeletal abnormalities and short stature in the affected individuals [15]. Another candidate is FGF16, which is a member of fibroblast growth factor family shown to play various roles in developmental processes including morphogenesis. Mouse studies proposed a crucial role for FGF16 in cardiac morphogenesis [20] and, recently, nonsense mutations in FGF16 were demonstrated to associate with congenital limb malformations [21], suggesting the involvement of the gene in human skeletal development. In the second height locus, ITM2A is a functional candidate: The association signal maps just 35 kb upstream of the transcription start site of ITM2A, a gene known to be involved in cartilage development [13,14]. Additionally, our eQTL analysis provided further evidence showing that the height-associated variants in this locus also influence the expression of ITM2A. Interestingly, high expression of ITM2A in adipose tissue stem cells has been proposed to inhibit the initiation of chondrogenesis in these cells [22]. As the allele associated with shorter stature associated with increased expression of ITM2A, this suggests the allelic effect to height could be mediated through the capacity to generate cartilage and bone.

Given the unique nature of chrX compared to autosomes, i.e., males being hemizygous for the chromosome and the partly incomplete XCI, which could give rise to sexual dimorphisms, we analyzed our association data also by accounting for potential heterogeneity in genetic effects between sexes. This led to the discovery of considerable sex-difference in the genetic effects for height near ITM2A, while for the other two associated loci there was no such evidence. A large-scale scan on 270,000 individuals for sexual dimorphisms in autosomal genetic effects for various anthropometric phenotypes identified significant sex differences only in associations for waist phenotypes and none was observed for height [23]. Additionally, the few previous height GWAS that had significant or suggestive discoveries in chrX [17-19] did not, to our knowledge, evaluate for potential sex heterogeneity in the chrX associations. Therefore, the sex difference in genetic effects for height in the ITM2A locus here is likely the first demonstration of its kind and hence warrants the investigation for potential sex heterogeneity in associations for other X-linked loci.

The difference in the genetic effects for males and females near ITM2A appeared fully consistent with no dosage compensation in this locus. While we cannot completely exclude the possibility that such a difference can also arise through some other type of sex-by-SNP interaction effect, we gained further confirmation for the lack of dosage compensation for height in ITM2A by quantifying the evidence for the two dosage compensation models in each of the associated loci using a new model comparison framework. Additionally, the gene expression data that showed higher expression of ITM2A in women, as expected if two X chromosomes are transcriptionally active in this locus. Similarly, expression of ITM2A was previously observed to be female-biased in monocytes and in the same study the cis-eQTL for ITM2A showed sex heterogeneity [24]. Furthermore, strongly speaking for the role of impartial silencing of this locus, in a comprehensive survey into XCI, ITM2A was found to be among the 10% of chrX genes, which variably escape from inactivation, i.e., are expressed from the inactivated copy of X (Xi) in a majority of women [8]. For comparison, the expression of ATRX and MAGT1 in the other height locus (FGF16 was not included in the survey) from Xi is fully silenced [8].

Given the converging evidence from our association study, statistical model comparison of the association data, gene expression data and literature, it seems likely that our observations of sex heterogeneity in the genetic effects for height near ITM2A are due to incomplete dosage compensation in this locus. Therefore, our study likely provides, to the best of our knowledge, the first link between an XCI-escaping gene and phenotypic variation in non-syndromic individuals. This discovery has several plausible implications. As increased expression of ITM2A links with shorter stature, the greater dosage of ITM2A in women compared to men may explain some of the substantial sex difference observed in height. In addition to the sex differences in the overall expression levels, incomplete dosage compensation also causes the genetic variation in the population to have different effects on the trait distribution of males and females. Assuming that rs1751138, the ITM2A lead SNP, is the causal variant for the observed height association, we estimate that the observed 36% frequency of the height decreasing allele accounts for 1.5% of the current difference in mean height between men and women in the Finnish population, when compared to a population that was monomorphic for the major allele at this SNP (Materials and Methods). Besides contributing to sexual dimorphisms, XCI-escaping genes are candidates for causing the abnormalities in chrX aneuploidies. Hence, our findings also highlight *ITM2A* as a potential contributor to the height phenotype often observed in individuals with an unusual number of X chromosomes [25].

To conclude, our findings illustrate the value of including the chrX in large-scale genetic studies and provide a motivation to assess the chrX associations in larger sample sizes, particularly for traits where we estimated chrX to explain part of the trait variation. We anticipate that such studies will identify further loci that contribute to the heritability of complex traits, as well as increase our understanding of their genetic architecture and underlying biology. As evidenced by our observations in the *ITM2A* locus, studying chrX association opens avenues for the discovery of links between phenotypes and loci that escape from XCI. Such associations bear potential to bring insights into the biological bases of sexually dimorphic traits such as complex diseases with different incidences between males and females.

Materials and Methods

Cohorts

The study included individuals from seven discovery cohorts: The Northern Finland Birth Cohort 1966 (NFBC1966), The Cardiovascular Risk in Young Finns Study (YFS), The COR-OGENE Study (COROGENE), Helsinki Birth Cohort Study (HBCS), the Health 2000 GenMets Study (GenMets), the Diabetes Genetics Initiative (DGI) and a prospective cardiovascular disease (CVD) case-control sample from the FINRISK collections (PredictCVD). In the replication stage a further subset of individuals from the FINRISK collections (FR) was included. A summary of the cohort characteristics is given in Table 1. All participants gave an informed consent and the data was deidentified for all analyses.

NFBC1966 is a birth cohort study of children born in 1966 in the two northernmost provinces of Finland originally designed to focus on factors affecting pre-term birth, low birth weight, and subsequent morbidity and mortality [26]. The blood sample for the DNA extraction and all phenotype data (except the childhood height measurements) used in the present study were collected at a follow-up visit when the participants were 31 years of age. The COROGENE cohort includes acute coronary syndrome patients who underwent coronary angiogram between June 2006 and March 2008 in the Helsinki University Central Hospital [27] and matched controls from the Helsinki-Vantaa region participants of FINRISK 1997, 2002, and 2007 surveys performed in Finland every five years since 1972 [28]. In the current study, the COROGENE cases were only included in the analyses of height and body-mass-index. The DGI sample consists of individuals from Sweden and Finland and was originally designed to identify loci associated with type 2 diabetes (T2D) [29]. The sample consists of patients with T2D; gender, BMI, age and geographically matched controls and discordant sibships. The current analysis was performed on 3142 individuals, including all individuals from the original analysis and individuals previously excluded after having been identified as belonging to pairs of samples identified as cryptic first degree relatives [29]. GenMets is a subset from the Health 2000 survey collected in 2000-2001 to obtain information on the most important public health problems in Finland [30]. The cohort includes metabolic syndrome cases and their matched controls aged 30 years and above. YFS is a longitudinal follow-up study of children and adolescents from all around Finland initiated in 1980 to study the cardiovascular risk

from childhood to adulthood [31], and the data for the present study is from the 27-year follow-up when the participants were 30– 45 years of age. The PredictCVD study comprises of incident cardiovascular disease cases and matched controls selected from the 1992, 1997, 2002 and 2007 FINRISK surveys [28]. For the analyses of the current study, the 96 individuals included and genotyped in both COROGENE and PredictCVD samples were excluded from the COROGENE data set. The FR sample used for replication includes a random, yet non-overlapping with COROGENE and PredictCVD, subset of individuals from FINRISK surveys from 1997 and 2002.

Genotyping and imputation

The following genotyping arrays were used for genotyping the study cohorts: Illumina 370K array for NFBC1966, Illumina 610K array for COROGENE and GenMets, custom generated Illumina 670K array for YFS and HBCS, Illumina 770K array for PredictCVD, Illumina HumanCoreExome-12v1-0 for FR and Affymetrix GeneChip Human Mapping 500K Array set for DGI. The quality control procedures included removing closely related individuals (PI_HAT >0.1) by analyzing pairwise IBD relationships for all individuals in five Finnish discovery cohorts together and have been described previously in detail [29,32]. The imputation of non-pseudoautosomal chrX variants into the study cohorts was performed on the cleaned data in each cohort using IMPUTE version 2.2.2 [33,34]. The reference panel used in the imputation was the integrated variant set release (v3) released in March 2012 (http://mathgen.stats.ox.ac.uk/impute/ data_download_1000G_phase1_integrated.html). The data were split into genomic regions of ~ 5 Mb (with 250 kb (DGI) or 1 Mb (other cohorts) buffer region), using effective population size of 20000 (DGI) or 11418 and k value of 80.

Phenotype adjustments

Within each cohort, all twelve phenotypes were adjusted for males and females separately using age (not in NFBC1966) and ten first principal components as covariates. Additionally, WHR and blood pressure measurements were adjusted for BMI. The residuals from the linear regression were inverse normal transformed to have mean 0 and standard deviation 1, and the normalized residuals were then used as phenotypes in the variance estimate and association analyses. In cohorts where the information was available, individuals on lipid-lowering medication were excluded prior the covariate adjustment for the blood lipids (TC, TG, LDL-C and HDL-C), blood-pressure medication was an exclusion criterion for systolic and diastolic blood pressure and diabetes medication for glucose and insulin. Non-fasting individuals were excluded from the analysis of glucose and insulin, and pregnant women were only included in the analysis of CRP and height. All phenotype adjustments and data normalization were done in STATA/SE 12.1 or R (version 2.11.1). The sample size for each phenotype in the discovery cohorts, i.e., the number of individuals with both genotype and phenotype information, are given in Table S5.

Estimation of the variance explained by the X chromosome and the autosomes

We estimated how much phenotypic variance a panel of 217,112 high quality SNPs from chrX (IMPUTE2 info >0.8, MAF >0.01) explain using the linear mixed model approach implemented in GCTA (v.1.13) [10]. This analysis included six of the seven discovery cohorts (NFBC1966, COROGENE, Gen-Mets, YFS, HBCS and PredictCVD) for which we had access to

the individual genotype data. Following the work of Yang et al. [2], we applied three models for dosage compensation: full dosage compensation (FDC), no dosage compensation (NDC) and equal variance in both sexes (EV). However, none of the models was consistently favored over the other two across the traits (Table S1). In the main text we report the variance estimates from the EV model because those best capture the average genetic contribution of chrX to the population: in FDC and NDC models the variances between males and females are different and for both models the mean of the sex-specific variances is close to the single value given by the EV model in our data. Furthermore, since the traits have been normalized to have a variance of 1 within each sex, the models that assume different genetic variance between the sexes (namely FDC and NDC) should also allow different residual variances between the sexes but this has not been implemented in GCTA (v.1.13). We report the results from all three models using imputed SNPs in Table S1.

For a comparison, we also estimated the genetic variances using 9,517 directly genotyped SNPs in chrX but did not find notable differences from the results with the imputed data (Table S6). For another comparison, we estimated the variance explained by the 319,445 directly genotyped SNPs with MAF>0.01 in the autosomes (Table 2). If the variants contributing to the traits were uniformly distributed across the genome, then we would expect that chrX genetic variance is about 3% of the autosomal genetic variance, as about 3% of the genetic variation in our data is in the X chromosome. Figure S3 plots estimated autosomal genetic variance against chrX one.

All mixed model analyses excluded individuals in such a way that none of the remaining pairs of individuals had an estimated relatedness coefficient r >0.05 and the same trait values were used as with the association analyses. For comparison, the analyses were also carried out by using r = 0.025 as the relatedness cut-off (Table S7).

Chromosome X-wide association analyses

The associations between variants and phenotypes were tested for all available genotyped or imputed SNPs in chrX separately for men and women in each cohort encoding genotypes $\{0,2\}$ in men and $\{0,1,2\}$ in women, i.e., assuming that one of the two X chromosomes in women is fully inactivated. In DGI the analysis was performed using a linear mixed model that can account for sample structure, as implemented in EMMAX [35]. In other cohorts the analyses were performed in SNPTEST [11] (version 2.4.0) assuming an additive genetic model and using expected genotype counts. In each cohort the results were filtered prior the meta-analysis to include only good-quality variants (SNPTEST info in women >0.4) and variants with more than three copies of the minor allele (minor allele count in women >3) resulting in between 323,564 and 383,337 variants per cohort in the discovery set.

The cohorts and genders were combined in a fixed-effects (inverse variance weighted) meta-analysis and in a sex-differentiated meta-analysis, which combines the results allowing for the allelic effects to differ between men and women and also conducts a meta-analysis separately for the results from men and from women, both meta-analysis options implemented in GWAMA [12]. Genomic control was applied for the association results of each study to account for P-value inflation arising from residual population structure in the data or other confounding factors. The meta-analysis summary was also corrected with the genomic control lambda for those phenotypes where there was indication of inflation (lambda >1.0): The genomic control lambdas for the twelve phenotypes were between 0.94–1.13 in the discovery analysis (Table S8). For each phenotype the meta-analysis results were further filtered to include only the SNPs for which the association result was available from at least two of the input files, hence association results were available for up to 405,411 SNPs. Adopting the genome-wide significance P-value threshold, P-value< 5×10^{-8} , three loci showed significant associations in the discovery analysis. In these associated loci the lead SNPs were imputed with high quality (imputation info >0.80) in all cohorts.

Height associations in childhood

To investigate whether the observed associations with height near ITM2A and ATRX are present already in pre-puberty, the association of the chrX SNPs with childhood height was studied in a subset of individuals from NFBC1966 and YFS cohorts (N(males) = 1204)and N(females) = 1189in NFBC1966; N(males) = 417 and N(females) = 478 in YFS) who had height measurements available both from pre-puberty (ages 8-10 years for NFBC1966 and 9 years for YFS) and adulthood. The height measurements were adjusted for age (in NFBC1966) and ten first principal components, the residuals were inverse normal transformed and subsequently the association of the SNP dosage with the transformed residuals was studied in SNPTEST using an additive model of association. The SNP-phenotype associations for the two height measurements were studied in both cohorts, separately for the genders, and the results were combined in a fixed-effects and a sex-differentiated meta-analysis using GWAMA applying genomic control correction. The association results for the height lead SNPs with childhood and adulthood height in these individuals are given in Table S4.

Cis-eQTL analysis of the associated regions

A subset of the COROGENE cohort (N=513) had both genome-wide SNP data and gene expression data from whole blood, assayed using Illumina HumanHT-12v3 Expression BeadChips, available, as described previously [36]. The associations between the three lead SNPs and all gene expression probes within 1 Mb of the SNPs was studied in SNPTEST using an additive model of association. For the one significantly associated expression probe (*ITM2A*/ILMN_2076600) we further evaluated whether the level of expression was different between the sexes by comparing the unadjusted expression in males and females using Student's t-test in R.

Variance explained by individual loci in chromosome X

As males carry only one copy of the X chromosome, the genotype variances of the SNPs in the non-pseudoautosomal region of chrX differ between women and men. Assuming the model of complete inactivation of one the X chromosomes in women, and hence coding the X-linked alleles $\{0,2\}$ in men and $\{0,1,2\}$ in women, the genotype variance in men is twice that in women: 2P(1-P) for females and 4P(1-P) males, where P denotes the allele frequency of the SNP. Thus, the estimation of the variance explained from the meta-analysis summary statistics should be evaluated for the genders separately using the formulas $2P(1-P)b_F^2$ for women and $4P(1-P)b_M^2$ for men, where b_F and b_M denote the standardized effect sizes in women and men, respectively.

Comparison of dosage compensation models

We applied a Bayesian framework [37] to compare full dosage compensation (FDC) and no dosage compensation (NDC) models at the top SNPs of the three associated regions. We used the estimated effect sizes b_F (allelic effect in females)

and b_M (effects in males when two genotypes are coded 0 and 2) together with their standard errors to approximate the likelihood function as in [16]. The two models differ in their prior specification:

FDC: $b_F \sim N(0, s^2)$, $b_M \sim N(0, s^2)$ and $cor(b_F, b_M) = 1$

NDC: $b_F \sim N(0, s^2)$, $b_M \sim N(0, 0.25 s^2)$ and $cor(b_F, b_M) = 1$

Where s^2 , the variance of the prior effect size in females, depends on the allele frequency of the SNP and is chosen so that with 95% probability the studied SNP explains less than 1% of the variance of the trait. Intuitively, according to the FDC model $b_F = b_M$ whereas according to the NDC model $b_F = 2b_M$. Bayes factors between the two models can be computed using the approximate likelihood approach [16] and the posterior probabilities of the NDC model are shown in Figure 2 under the assumption that each model is equally likely *a priori*.

Effect of lack of dosage compensation on sex difference in mean trait value

Suppose that each copy of the minor allele ('a' with frequency f_a) at a particular SNP on chrX causally affects the trait value by 'b' both in males and in females, i.e., there is no dosage compensation. To simplify notation, we assume that the mean trait value of males with genotype 'A' is the same as the mean trait value in females with genotype 'AA', and denote this mean by 'm'. The overall mean trait values in males, 'm_M', and in females, 'm_F', are $m_M = (1 - f_a) \cdot m + f_a \cdot (m + b) = m + f_a \cdot b$, and

 $\mathbf{m}_{\mathrm{F}} = \mathbf{f}_{\mathrm{AA}} \bullet \mathbf{m} \ + \ \mathbf{f}_{\mathrm{Aa}} \bullet (\mathbf{m} + \mathbf{b}) \ + \ \mathbf{f}_{\mathrm{aa}} \bullet (\mathbf{m} + 2\mathbf{b}) = \mathbf{m} \ + \ (\mathbf{f}_{\mathrm{Aa}} + 2\mathbf{f}_{\mathrm{aa}}) \bullet \mathbf{b} = \mathbf{m} \ + \ 2\mathbf{f}_{\mathrm{a}} \bullet \mathbf{b}$

The difference in the sex-specific means is

 $m_M - m_F = m + f_a \cdot b - (m + 2f_a \cdot b) = -f_a \cdot b$

So the effect of allele 'a' is either to increase (if b < 0) or decrease (if b > 0) the male-female difference in the mean trait value by $|f_a \cdot b|$, compared to the situation where only allele 'A' was present in the population.

Application to rs1751138, the lead SNP of the association near *ITM2A*:

b = -0.555 cm (se = 0.0734 cm). This is a fixed-effects estimate of the allelic effects in quantile normalized height in females -0.092559 (0.013934) and males -0.071802 (0.019274), multiplied by an estimate of the standard deviation of height in Finland, 6.5 cm.

Thus by introducing the minor allele (A) with frequency $f_a = 0.36$ in the population, the male-female difference in mean height increases by 0.36*0.555 cm = 0.20 cm. As the mean difference in height between the sexes in Finland is about 13.7 cm, the variation at this SNP accounts for 0.20 cm/ 13.7 cm = $0.0146 \approx 1.5\%$ of that difference.

Supporting Information

Figure S1 Regional association plots of the associations in the two height loci after conditioning the association analysis on rs1751138, the lead associated SNP in the *ITM2A* height locus. A: The region near *FGF16*, *ATRX* and *MAGT1*. The lead associated SNP in this analysis, rs34979608, is highlighted with diamonds, the yellow diamond indicating the association in the discovery analysis and the purple diamond and text the association in the joint analysis of discovery and replication cohorts. The yellow × indicates the association of the SNP that was most associated with height in this region in the discovery meta-analysis, i.e., before the conditional analysis (P-values 3.02×10^{-7} and 2.80×10^{-8} in the conditional analysis using the discovery cohorts and both discovery and replication data, respectively). B: The region near *ITM2A*. No association with height remains in this region in the

conditional analysis. Each circle in the plots indicates a SNP and the color of the circle shows the linkage disequilibrium, r^2 , of the SNP to the lead associated SNP in each region, rs34979608 in A and rs1751138 in B: dark blue ($r^2 < 0.2$), light blue ($r^2 > 0.2$), green ($r^2 > 0.4$), orange ($r^2 > 0.6$) and red ($r^2 > 0.8$). The correlation structure between the SNPs was calculated from Finnish data using the genotypes from the COROGENE cohort. The light blue line in the background and the right hand y-axis show the recombination rate in the region as calculated from HapMap CEU data.

(PDF)

Figure S2 Boxplots of *ITM2A* expression in men and women. The levels of whole blood *ITM2A* expression visualized using boxplots separately for men (blue) and women (orange) in the individuals of COROGENE cohort (N = 513) for whom expression data was available. The mean expression level is higher in women (P-value = 0.00251) providing support for incomplete dosage compensation between men and women in the *ITM2A* locus.

(PDF)

Figure S3 Comparison of the phenotypic variances attributable to chromosome X variants (y-axis) and autosomal variants (x-axis). The dashed line is y = 0.03x and shows the region of the expected values of the points under the assumption that the genetic effects are small and uniformly distributed across the genome and that the X chromosome contains about 3% of all genetic variation. The dotted lines show the standard errors of the estimates. The traits are TC: total cholesterol; LDL-C: low-density lipoprotein cholesterol; TG: total triglycerides; CRP: C-reactive protein; BMI: body-mass-index; WHR: waist-hip-ratio; SBP: systolic blood pressure; DBP: diastolic blood pressure; HGT: height; GLK: fasting glucose; INS: fasting insulin. (PNG)

Table S1 Estimates of the phenotypic variance attributable to 217,112 imputed and genotyped chrX SNPs using three different dosage compensation models.

(XLS)

Table S2 Association results in each discovery cohort for the lead SNPs in the three associated loci (rs1751138 near *ITM2A* and

References

- Hindorff LA, MacArthur JEBI, Morales JEBI, Junkins HA, Hall PN, et al. (2013) A Catalog of Published Genome-Wide Association Studies. Available: www. genome.gov/gwastudies
- Yang J, Manolio TA, Pasquale LR, Boerwinkle E, Caporaso N, et al. (2011) Genome partitioning of genetic variation for complex traits using common SNPs. Nat Genet 43: 519–525.
- Wise AL, Gyi L, Manolio TA (2013) eXclusion: Toward Integrating the X Chromosome in Genome-wide Association Analyses. Am J Hum Genet 92: 643– 647.
- Coviello AD, Haring R, Wellons M, Vaidya D, Lehtimaki T, et al. (2012) A genome-wide association meta-analysis of circulating sex hormone-binding globulin reveals multiple Loci implicated in sex steroid hormone regulation. PLoS Genet 8: e1002805.
- Chu X, Shen M, Xie F, Miao XJ, Shou WH, et al. (2013) An X chromosomewide association analysis identifies variants in GPR174 as a risk factor for Graves' disease. J Med Genet 50: 479–485.
- Heid IM, Jackson AU, Randall JC, Winkler TW, Qi L, et al. (2010) Metaanalysis identifies 13 new loci associated with waist-hip ratio and reveals sexual dimorphism in the genetic basis of fat distribution. Nat Genet 42: 949– 960.
- Teslovich TM, Musunuru K, Smith AV, Edmondson AC, Stylianou IM, et al. (2010) Biological, clinical and population relevance of 95 loci for blood lipids. Nature 466: 707–713.
- Carrel L, Willard HF (2005) X-inactivation profile reveals extensive variability in X-linked gene expression in females. Nature 434: 400–404.

rs 182838724 near FGF16, ATRX and MAGT1 for height and rs 139163435 in Xq23 for fasting insulin). (XLS)

Table S3 Lead associations in the height locus near *FGF16*, *ATRX* and *MAGT1* after conditioning the association analysis on the lead SNP of the *ITM2A* locus, rs1751138. (XLS)

Table S4 Associations of the most associated SNPs in the two height loci with height in childhood (age 8–10) and adulthood in a subset of the study subjects from two cohorts (NFBC1966 and YFS) for whom childhood height measurements were available. (XLS)

 Table S5
 Sample sizes for the twelve phenotypes in the discovery cohorts.

(XLS)

Table S6 Estimates of the phenotypic variance attributable to9,517 directly genotyped chrX SNPs using three different dosagecompensation models.(XLS)

Table S7 Estimates of the phenotypic variance attributable to 217,112 imputed and genotyped chrX SNPs using three different dosage compensation models using a threshold of 0.025 to remove related individuals.

(XLS)

Table S8 Genomic control lambdas for the twelve phenotypes in the X chromosome-wide meta-analyses of the discovery cohorts. In cases where lambda >1, the meta-analysis results were corrected using the lambda statistic. (XLS)

Acknowledgments

We thank C. Spencer for ideas about Bayesian model comparison and D.G. MacArthur for valuable comments on the manuscript.

Author Contributions

Conceived and designed the experiments: TT MPi JK SR. Analyzed the data: TT MPi APS CL JK. Contributed reagents/materials/analysis tools: TT MPi APS CL JK TL MLL MPe JS EV JGE LG AJ MRJ OTR VS SR. Wrote the paper: TT MPi SR.

- Genomes Project C, Abecasis GR, Auton A, Brooks LD, DePristo MA, et al. (2012) An integrated map of genetic variation from 1,092 human genomes. Nature 491: 56–65.
- Yang J, Lee SH, Goddard ME, Visscher PM (2011) GCTA: a tool for genomewide complex trait analysis. Am J Hum Genet 88: 76–82.
- Marchini J, Howie B (2010) Genotype imputation for genome-wide association studies. Nat Rev Genet 11: 499–511.
- Magi R, Lindgren CM, Morris AP (2010) Meta-analysis of sex-specific genomewide association studies. Genet Epidemiol 34: 846–853.
- Tuckermann JP, Pittois K, Partridge NC, Merregaert J, Angel P (2000) Collagenase-3 (MMP-13) and integral membrane protein 2a (Itm2a) are marker genes of chondrogenic/osteoblastic cells in bone formation: sequential temporal, and spatial expression of Itm2a, alkaline phosphatase, MMP-13, and osteocalcin in the mouse. J Bone Miner Res 15: 1257–1265.
- Van den Plas D, Merregaert J (2004) In vitro studies on Itm2a reveal its involvement in early stages of the chondrogenic differentiation pathway. Biol Cell 96: 463–470.
- 15. Gibbons R (2006) Alpha thalassa
emia-mental retardation, X linked. Orphanet J Rare Dis 1: 15.
- Band G, Le Q, Jostins L, Pirinen M, Kivinen K, et al. (2013) Imputation-Based Meta-Analysis of Severe Malaria in Three African Populations. PLoS Genet 9: e1003509.
- N'Diaye A, Chen GK, Palmer CD, Ge B, Tayo B, et al. (2011) Identification, replication, and fine-mapping of Loci associated with adult height in individuals of african ancestry. PLoS Genet 7: e1002298.

- Carty CL, Johnson NA, Hutter CM, Reiner AP, Peters U, et al. (2012) Genomewide association study of body height in African Americans: the Women's Health Initiative SNP Health Association Resource (SHARe). Hum Mol Genet 21: 711–720.
- Gudbjartsson DF, Walters GB, Thorleifsson G, Stefansson H, Halldorsson BV, et al. (2008) Many sequence variants affecting diversity of adult human height. Nat Genet 40: 609–615.
- Lu SY, Sheikh F, Sheppard PC, Fresnoza A, Duckworth ML, et al. (2008) FGF-16 is required for embryonic heart development. Biochem Biophys Res Commun 373: 270–274.
- Jamsheer A, Zemojtel T, Kolanczyk M, Stricker S, Hecht J, et al. (2013) Whole exome sequencing identifies FGF16 nonsense mutations as the cause of X-linked recessive metacarpal 4/5 fusion. J Med Genet 50: 579–584.
- Boeuf S, Borger M, Hennig T, Winter A, Kasten P, et al. (2009) Enhanced ITM2A expression inhibits chondrogenic differentiation of mesenchymal stem cells. Differentiation 78: 108–115.
- Randall JC, Winkler TW, Kutalik Z, Berndt SI, Jackson AU, et al. (2013) Sexstratified genome-wide association studies including 270,000 individuals show sexual dimorphism in genetic loci for anthropometric traits. PLoS Genet 9: e1003500.
- Castagne R, Zeller T, Rotival M, Szymczak S, Truong V, et al. (2011) Influence of sex and genetic variability on expression of X-linked genes in human monocytes. Genomics 98: 320–326.
- Ottesen AM, Aksglaede L, Garn I, Tartaglia N, Tassone F, et al. (2010) Increased number of sex chromosomes affects height in a nonlinear fashion: a study of 305 patients with sex chromosome aneuploidy. Am J Med Genet A 152A: 1206–1212.
- Rantakallio P (1969) Groups at risk in low birth weight infants and perinatal mortality. Acta Paediatr Scand 193: Suppl 193:191+.
- Vaara S, Nieminen MS, Lokki ML, Perola M, Pussinen PJ, et al. (2012) Cohort Profile: the Corogene study. Int J Epidemiol 41: 1265–1271.

- New Loci for Height and Fasting Insulin in chrX
- Vartiainen E, Laatikainen T, Peltonen M, Juolevi A, Mannisto S, et al. (2010) Thirty-five-year trends in cardiovascular risk factors in Finland. Int J Epidemiol 39: 504–518.
- Diabetes Genetics Initiative of Broad Institute of H, Mit LU, Novartis Institutes of BioMedical R, Saxena R, Voight BF, et al. (2007) Genome-wide association analysis identifies loci for type 2 diabetes and triglyceride levels. Science 316: 1331–1336.
- Perttila J, Merikanto K, Naukkarinen J, Surakka I, Martin NW, et al. (2009) OSBPL10, a novel candidate gene for high triglyceride trait in dyslipidemic Finnish subjects, regulates cellular lipid metabolism. J Mol Med (Berl) 87: 825– 835.
- Raitakari OT, Juonala M, Ronnemaa T, Keltikangas-Jarvinen L, Rasanen L, et al. (2008) Cohort profile: the cardiovascular risk in Young Finns Study. Int J Epidemiol 37: 1220–1226.
- Kettunen J, Tukiainen T, Sarin AP, Ortega-Alonso A, Tikkanen E, et al. (2012) Genome-wide association study identifies multiple loci influencing human serum metabolite levels. Nat Genet 44: 269–276.
- Howie BN, Donnelly P, Marchini J (2009) A flexible and accurate genotype imputation method for the next generation of genome-wide association studies. PLoS Genet 5: e1000529.
- Howie B, Marchini J, Stephens M (2011) Genotype imputation with thousands of genomes. G3 (Bethesda) 1: 457–470.
- Kang HM, Sul JH, Service SK, Zaitlen NA, Kong SY, et al. (2010) Variance component model to account for sample structure in genome-wide association studies. Nat Genet 42: 348–354.
- Inouye M, Silander K, Hamalainen E, Salomaa V, Harald K, et al. (2010) An immune response network associated with blood lipid levels. PLoS Genet 6: e1001113.
- International Stroke Genetics C, Wellcome Trust Case Control C, Bellenguez C, Bevan S, Gschwendtner A, et al. (2012) Genome-wide association study identifies a variant in HDAC9 associated with large vessel ischemic stroke. Nat Genet 44: 328–333.